Author’s response to reviews

Title: A modified M-stage classification based on the metastatic patterns of pancreatic neuroendocrine neoplasms: a population-based study

Authors:

Xianbin Zhang (zhangxianbin@hotmail.com)
Jiaxian Song (jiaxian-song@hotmail.com)
Peng Liu (surgeonliupeng@126.com)
Mohammad Mazid (dr_mazid14@hotmail.com)
Lili Lu (lulili_92@sina.com)
Yuru Shang (shangyuru@hotmail.com)
Yushan Wei (weiyushan@outlook.com)
Wenli Zhang (zhangwenli@dlmedu.edu.cn)
Houli Zhang (houlizh@163.com)
Peng Gong (gongpengdalian@163.com)
Li Ma (mali_lele@sina.com)

Version: 2 Date: 26 Sep 2018

Author’s response to reviews:

We have revised the manuscript according to the editor comments.

There is no comment of reviewers.